Author's response to reviews

Title: The Nigerian national blindness and visual impairment survey: rationale, objectives and detailed methodology

Authors:

Dineen B (brendan.dineen@nuigalway.ie)
Gilbert CE (Clare.Gilbert@lshtm.ac.uk)
Rabiu MM (mrabiu@hotmail.com)
Kyari F (fatimaygk@yahoo.co.uk)
Mahdi MM (drtafida@yahoo.com)
Abubakar T (abdullmm@yahoo.com)
Christian E (cccezelum@yahoo.com)
Gabriel E (entekume_66@yahoo.com)
Elhassan E (elizabethelhassan@yahoo.co.uk)
Abiose A (adenikeabioseo@yahoo.com)
Faal H (hfaal@sightsavers.org)
Jiya J (jiyajy@yahoo.com)
Ozemela C (cpozems@yahoo.co.uk)
Lee PS (pak.lee@ucl.ac.uk)
Murthy V S Gudlavalleti (gvsmurthy2000@yahoo.com)

Version: 5 Date: 1 September 2008

Author's response to reviews:

Dear Editors,

I have made a few corrections in the manuscript with regard to typographical errors and a few content changes and am submitting the revision.

regards,

Murthy Gudlavalleti